Author’s response to reviews

Title: Predictive abilities of cardiovascular biomarkers to rapid decline of renal function in Chinese community-dwelling population: a 5-year prospective analysis

Authors:
Shihui Fu (xiaoxiao0915@126.com)
Chunling Liu (442815017@qq.com)
Leiming Luo (lleim@sina.com)
Ping Ye (sci301@126.com)

Version: 2 Date: 01 Oct 2017

Author’s response to reviews:

BMC Nephrology
September 28, 2017
Response letter

Dear Editors and Reviewers:

On behalf of my co-authors, we thank you very much for giving us an opportunity to revise our manuscript. We appreciate the editors and reviewers very much for their positive and constructive comments and suggestions on our manuscript entitled "Predictive abilities of cardiovascular biomarkers to rapid decline of renal function in Chinese community-dwelling population: a 5-year prospective analysis”.

Those comments and suggestions are all valuable and very helpful for us to revise and improve our paper, as well as very important to direct our researches. We study the editors and reviewer's comments carefully and try our best to revise the paper. Attached please find the revised version, which we would like to submit for your kind consideration.

Responds to the editors and reviewers’ comments:
Editors:

1. Please include the sentence "All authors have read and approved the final version of this manuscript" in the Author contributions section and ensure that this is the case.

We are very sorry for the unclear description. Thank you for your valuable suggestion very much. Your valuable suggestions are very valuable and helpful for us to improve our paper. Under your guidance, we make the corresponding modification in the manuscript.

2. Please upload a clean version of the manuscript without tracked changes or highlighting and then remove any highlighted versions from the inventory, as these are no longer required at this stage of the editorial process.

We are very sorry for the unclear description. Thank you for your kind suggestion very much. Your valuable suggestions are very valuable and helpful for us to improve our paper. Under your guidance, we make the corresponding modification in the manuscript.

Reviewer 1

1. Please include all comments for the authors in this box rather than uploading your report as an attachment. Please only upload as attachments annotated versions of manuscripts, graphs, supporting materials or other aspects of your report which cannot be included in a text format.

We are very sorry for the unclear description. Thank you for your valuable suggestion very much. Your valuable suggestions are very valuable and helpful for us to improve our paper. Under your guidance, we make the corresponding modification in the manuscript.

We have to apologize for giving you so many troubles. We deeply appreciate your consideration and suggestions of our manuscript and look forward to receiving your comments. Your comments and suggestions give us not only the great help in revising the article, but also the significant revelation in our scientific research. Your kind guidance is our good luck. We wonder if the modification could meet your requirements. If you have any queries, please don’t hesitate to contact us.

Thank you and best regards.